Author's response to reviews

Title: Solid Food Protein induced Enterocolitis Syndrome - four new cases, a brief review of the literature and possible new insights of the pathological mechanisms

Authors:

Amolak S Bansal (Amolak.Bansal@esth.nhs.uk)
Sree Bhaskaran (Sree.Bhaskaran@esth.nhs.uk)
Rhea A Bansal (RheaBansal@yahoo.co.uk)

Version: 3 Date: 21 January 2012

Author's response to reviews:

The four cases have now been described as required by the journal and we have removed the summary table as this duplicated the information in the text. Hopefully this revised version now meets all of the requirements of the journal.

Many thanks

Dr Amolak S Bansal